Math 9C–001 Winter 2006 R. Schultz

STATISTICS FROM THE FIRST IN–CLASS EXAMINATION

The cutoff scores for letter grades are as follows:

A — 85 B — 70 C — 50 D — 30

The median score was 71.

Course policy on grade inquiries and appeals

Appeals regarding the grading of this examination must be submitted either to one of the teaching assistants or the primary instructor by the end of class on Wednesday, February 15. Written comments should be placed on the examination indicating the problems to be reconsidered. BRIEF and OBJECTIVE statements about specific issues may be included.